\begin{table}%T5
%\centering
\small
%\tabcolsep0.10cm
\newcommand{\hrv}   {{\small A5{\rm III}}}
\newcommand{\hrvi}  {{\small A6{\rm III}}}
\newcommand{\hrvii} {{\small A7{\rm III}}}
\caption{\label{tbl-hrdata}
  Additional data of HR~5999. The \Vio-value marked with (\dag) is uncertain, as
  the brightness is underestimated due to the degradation of the FES counter
  after 1990 (Sect.~\ref{sec-fes}).
  See Table~\ref{tbl-afgstar} for the other input parameters.
  The apparently good agreement of the 3 faint spectra (\#10$-$12) at the long
  wavelength is caused by the fact that we used the same (overexposed) image
  LWR14027 in these cases.
}
\begin{tabular}{rl clccc r}\hline\hline
\# &\cViue  &\cIUE  &\\cs &\multicolumn{3}{c}{\AVcs \umag}&Fig. \\ \cline{5-7}\\[-3.2mm]
   &\umag   &images &-- &\hrv  &\hrvi &\hrvii &{(A6)}   \\ \kline 
 1 &6.72    &09-01  &3.10  &0.59  &0.58  &0.48   &\ref{fig-hr672}\\ %\mybx{326372}{106364}
 2 &6.83\dag&09-02  &3.50  &0.70  &0.69  &0.59   &\ref{fig-hr683}\\ %\mybx{342497}{121267}
 3 &6.81    &09-03  &3.75  &0.68  &0.67  &0.57   &                  %\mybx{309973}{121267*}
                                                  \ref{fig-sed-hr}, \ref{fig-hr681}\\
 4 &7.02    &09-04  &4.50  &0.89  &0.88  &0.78   &\ref{fig-hr702}\\ %\mybx{301569}{201511}
 5 &7.16    &09-05  &4.50  &1.03  &1.02  &0.92   &\ref{fig-hr716}\\ %\mybx{321020}{216768}
 6 &7.17    &09-06  &4.50  &1.04  &1.03  &0.93   &\ref{fig-hr717}\\ %\mybx{320861}{216706}
%%%%%%%     &09-..  &....  &....  &....  &....   &\\                %\mybx{320863}{216707}
 7 &7.20    &09-07  &4.50  &1.07  &1.06  &0.96   &\ref{fig-hr720}\\ %\mybx{328853}{216706*}
 8 &7.30    &09-08  &4.50  &1.17  &1.16  &1.06   &\ref{fig-hr730}\\ %\mybx{321276}{216970}
 9 &7.39    &09-09  &5.00  &1.26  &1.25  &1.15   &\ref{fig-hr739}\\ %\mybx{320133}{216970*}
10 &7.47    &09-10  &5.25  &1.34  &1.33  &1.23   &\ref{fig-hr747}\\ %\mybx{325390}{214027*}
11 &7.49    &09-11  &5.50  &1.36  &1.35  &1.25   &\ref{fig-hr749}\\ %\mybx{325379}{214027*}
12 &7.57    &09-12  &5.75  &1.44  &1.43  &1.33   &\ref{fig-hr757}\\ %\mybx{317790}{214027}
\kline \end{tabular}  \end{table}